Author's response to reviews

Title: Experiences in applying skills learned in a Mental Health First Aid training course: a qualitative study of participants' stories

Authors:

   Anthony F Jorm (ajorm@unimelb.edu.au)
   Betty A Kitchener (bettyk@unimelb.edu.au)
   Stephen K Mugford (stephen.mugford@qsr.com)

Version: 5 Date: 31 October 2005

Author's response to reviews: see over
We have carried out the following tasks as requested by the Editor:

1. Checked that the title and abstract are identical in the manuscript and in the submission system.
2. Checked for accuracy and consistency of the authors.
3. Removed the boxes from the Results and put the quotes in italics.
4. Tables 2 and 3 have been formatted as requested.

We have decided not to make changes based on the reviewer’s discretionary comments.